Author's response to reviews

Title: Plasma leptin and insulin-like growth factor I levels during acute exacerbations of Chronic Obstructive Pulmonary Disease

Authors:

Prokopis Kythreotis Dr (pkc@otenet.gr)
Aggeliki Kokkini Dr (pkc@otenet.gr)
Stavrina Avgeropoulou Dr (stavrinavg@yahoo.gr)
Argyro Hadjioannou Dr (arghadjioanou@yahoo.gr)
Efgenia Anastasakou Dr (efganastasakou@yahoo.gr)
Antonis Rasidakis Prof (rasid@ath.forthnet.gr)
Petros Bakakos Dr (petros44@hotmail.com)

Version: 2 Date: 16 September 2008

Author's response to reviews:

Dear editor,

We kindly submit our manuscript entitled "Plasma leptin and insulin-like growth factor I levels during acute exacerbations of Chronic Obstructive Pulmonary Disease' addresssing changes in plasma levels of leptin and IGF-I in this group of patients during exacerbations and possible consequences.

We are awaiting for your comments.

Kind regards,

Dr Petros Bakakos
Lecturer in Respiratory Medicine